Author's response to reviews

Title: Trends of Increase in Western Medical Services in Traditional Medicine Hospitals in China

Authors:

Jay J Shen (jay.shen@unlv.edu)
Ying Wang (wangying771029@yahoo.com.cn)
Fang Lin (fanglin@ou.edu)
Jun Lu (lujun@shmu.edu.cn)
Charles B Moseley (charles.moseley@unlv.edu)
Mei Sun (guaisunmei@yahoo.com.cn)
Mo Hao (haomo03@fudan.edu.cn)

Version: 3 Date: 27 August 2011

Author's response to reviews:

Title page: author list, affiliations, and email addresses have been added.
Competing interests: Have been added.
Authors' contributions: Have been added.
References: The format has been changed to 'text [1].
Corresponding authors: two corresponding authors have been indicated. Please be aware of the order the two corresponding authors - 1. Mo Hao and 2. Jay J. Shen
Acknowledgments: Have been moved the place below the authors' contributions.
Abstract: 'Research objective” has been changed to "Background".
Conclusion: "Conclusion" section has been added after the Discussion.